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Data Taking
Day Delivered Recorded Efficiency Comment

(%)

2 Oct 5.36 4.91 92

3 Oct 0.00 0.00 0 Controlled Access

4 Oct 0.57 0.51 89 Controlled Access

5 Oct 3.79 3.31 88 Special Muon runs

6 Oct 3.03 2.59 86 One hour down due to 1553 bus problem

7 Oct 4.55 4.06 89

8 Oct 3.20 2.57 80 Muon scintillator HV trips.

2-8 Oct 20.5 18.0 88

Lum (pb-1) Lum (pb-1)
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Accesses
● Tuesday 3 October

– 6 hour access at 0900

● AFE power supplies are occasionally being turned off.  Replaced Rack 
monitor and continued investigation.

●  Also worked on CTT, CAL, PDT, Muon pixels, and PDT cleaning.
● Wednesday 4 October

– 1.5 hour access at 1000

● Isolate 1553 bus for power supplies from AFE boards to attempt to locate 
signal that turned off power supplies.  Fix PDT front end board.

– One hour access at 1600

● Four AFE2 boards went into reset.
● We found that the voltage on a power supply was too high.  This was 

adjusted and other supplies were checked. 
– D0 thanks the AD for delaying the shot so that we could make this repair.
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